Author's response to reviews

Title: The key to managing the risk of cancer in Cowden Syndrome: a case report

Authors:

HAMMAMI Sonia (soniawhammami@yahoo.fr)
BERRICHE Olfa (olfaberriche@yahoo.fr)
BELHADJALI Hichem (Belhadjalihichem@yahoo.fr)
HELLARA Olfa (olfahellara@rns.tn)
Faroog A (afaroog@nhs.net)
MAHJOUB Sylvia (sylviamahjoub@yahoo.fr)

Version: 9 Date: 7 March 2012

Author's response to reviews: see over
Editor in Chief,
Journal of Medical Case reports

Dear Editor in Chief,

Please find enclosed the reviewed manuscript of our paper:” The key to managing the risk of cancer in Cowden Syndrome: a case report” co-authored by Sonia Hammami et al, Journal reference CE: 2071914211624960; which we would like to submit once more for publication in your Journal. It contains original results which have not been submitted for publication elsewhere. In which, and upon the reviewer comments, we have requested modifications, some language correction by a native English speaker, clarified some aspects and corrected the mistakes encountered in the first version to make in order this version as clear as possible.

You will find also ‘in attach’ in separated:

1/ Letter to the Editor in Chief

2/ Answers to reviewer’s comments on separate page,

3/ Revised manuscript under new form.

We thank you in advance for the consideration given this paper and look forward to a favourable response from you soon

Best regards

Dr Sonia Hammami

Department of Internal Medicine
CHU F Bourguiba Monastir (Tunisia)
Email Sonia.Hammami@fmm.rnu.tn